Author's response to reviews

Title: Sleep study, respiratory mechanics, chemosensitive response and quality of life in morbidly obese patients undergoing bariatric surgery: A prospective, randomized, controlled.

Authors:

- Luis V F Oliveira (oliveira.lvf@pq.cnpg.br)
- Isabella C Aguiar (bellaaguiar@yahoo.com.br)
- Raquel P Hirata (raquel_hirata@hotmail.com)
- Newton S Faria Junior (nsdfj@yahoo.com.br)
- Israel S Reis (ireissantos@uol.com.br)
- Luciana M M Sampaio (lucianamalosa@uninove.br)
- Claudia S Oliveira (csantos@uninove.br)
- Paulo T C Carvalho (ptpaulo@terra.com.br)
- Fernando S S Leitao Filho (fernandostudart@uol.com.br)
- Lilian C Giannasi (odontogiannasi@uol.com.br)
- Carlos Alberto Malheiros (camalheiros@terra.com.br)
- Wilson Rodrigues Freitas Jr (wrfjr@uol.com.br)

Version: 3 Date: 4 October 2011

Author's response to reviews: see over
Title: Sleep study, respiratory mechanics, chemosensitive response and quality of life in morbidly obese patients undergoing bariatric surgery: A prospective, randomized, controlled.
Journal: BMC Surgery
MS ID: 7363195258005093

The BMC Series Editorial Production Team

Major revisions

- Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is "First name Initial Last name" (e.g. Joe F Bloggs)

We have done the corrections regarding the list of authors in the manuscript and in the submission system, in both it is identical.

- Abstract: This should be identical in the manuscript and the submission system.

The abstract is identical in the manuscript and the submission system.

- References: Please make sure references are cited in numerical order. We notice reference 31 is not cited within the main text. Please make sure references continue to be cited in numerical order once this reference is cited.

The reference 31 is the reference of sample size. This is already inserted in the correct order in main text and references. All the references are now revised.

- Additional files: additional files not mentioned in the text should be removed from the system

We have already inserted the figure title and legends below the reference list.

- Figures: Please provide figure titles and legends. These should be part of the manuscript file after the reference list. The figures are numbered automatically in the order in which they are uploaded.

We have already inserted the figure title and legends below the reference list.
**Minor revisions**

- **Title:** please use sentence case in the title, both in the manuscript file and the submission system. Capitalize only the first word and proper nouns, and do not include full stops/periods. For example, "The Title of My Prestigious Publication: A Scientific Study." should be changed to "The title of my prestigious publication: a scientific study"

The title was modified as requested both in the manuscript and in the submission system.

- **Please remove "Keywords"**

The Keywords were removed.

- **If an article cites a table from another article please spell out the table number (for example, one, two) This will avoid confusion with tables that are included within the main text.**

This topic is not applied.

- **References:** each citation can contain up to 30 authors' names. The term "et al." should not be used if there are fewer than 30 authors.

The references were modified as requested.

- **Figures:** It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimize white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: [http://www.biomedcentral.com/info/ifora/figures](http://www.biomedcentral.com/info/ifora/figures)

The figure was modified as requested.

- **Typography:** Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

We have made the final corrections and revisions for any typographical errors.

Thank you for your attention.

Isabella Aguiar

Co-author